Author's response to reviews

Title: Extensive haemorrhagic necrosis of liver is an unpredictable fatal complication in dengue infection: a postmortem study

Authors:

S.A.M. Kularatne (samkul@slt.net.lk)
I.V.B. Imbulpitiya (dhin_im@yahoo.com)
R.A. Abeysekera (raijthaasa55@gmail.com)
R. Waduge (roshitha2006@yahoo.com)
R.P.V.J. Rajapakse (jayanthar@pdn.ac.lk)
K.G.A.D. Weerakoon (kosalamadw83@gmail.com)

Version: 6
Date: 26 February 2014

Author's response to reviews:

The editors
BMC infectious disease

Thank you for accepting this manuscript for publication in your esteemed journal. As requested we have included that "infomed written consent from the next of kin for publication of these individual clinical details and the accompanying images was obtained" at the end of the methodology. Hope this would be satisfactory.

Thank you.
Yours trustfully,
Dr. Rajitha Abeysekera (corresponding author)